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B nntepsaine temmneparyp 250—350 °C nsydueno Bin-
SHWEe TIPUPOJIbI U KOHIEHTPAINU MOAM(PUKATOPOB
(Mg, P) Ha TeKCTypHbIe, KUCIOTHBIE U KaTaJluTHiec-
KIe CBOWCTBA I[eoJTa THIa ZSM-5 B peaxInu ajiKi-
JINPOBaHMS TOJIyosIa u3onponanoysoM. Ha HeMonan-
prmmpoBarHOM KaTaymmsatope HZSM-5 ankmmpo-
BaHHe TpoTeKaeT HecesekTHBHO. IIpn 330—350 °C
conepkanne usonpormaronyona (4-NIIT) B cmecn
usonporurrosryonos (WIIT) cocrasnsier 34.3—44.2 %.
Oxcu/i Maraust UrpaeT BaXKHYIO POJib B CEJEKTUB-
HOCTH HPOJYKTOB PEAKIUU U B paclpejieseHnn
nzomepo UIIT. MoauduinupoBanme ImeoauTa
HZSM-5 marameM cnocoOCTBYET YBEJUYEHWIO Ce-
aekrusroct o WIIT mo 69.6% Mmac. u pesromy
yBesnuennto copepsxkanus 4-NUIIT B emecu UIIT no
70.8%. lonmoHuTebHOE MOANMUITIPOBAHIE KaTa-
sguzaropa 5% Mg-HZSM-5 ruapodocdarom aMmo-
HUS CYIIECTBEHHO BJIMSIET HA €r0 KATAJINTHYECKHUe
cBoiicTBa. Pacnpeesenne H30MpONUITOIYOIOB 32~
BUCHT OT COJIePsKaHus U OT KOHIleHTpaIun ¢pocdo-
pa B KarajusaTtope. YBejJudyeHHe KOHIEHTPAIUN
docdopa B cocrase 5% Mg-HZSM-5 0 3.0% Mac.
HPUBOJIUT K 3HAYUTETHHOMY CHHIKEHHMIO TOGOUHBIX
MPOAYKTOB, yBeJaudeHuio cejexruBHocTu mo UIIT
(73.5-76.0 %), a TaKk:Ke YBEIMYEHUIO COIEPIKAHMS
4-UTIT B cmecu UTIT go 76.2%. [lanbHeiinee yBesi-
yenue cojepskatns docdopa B 5% Mg-HZSM-5 3a-
METHO CHIKaeT ceslekTnBHocTb 1o MITIT (69.6—72.1 %).
YcTaHoBIEHO, YTO C yBeJUYeHHEM COJEPIKAHUs
MOAU(PUKATOPOB MPOUCXOJUT YMEHDIIEHNE Y/1eb-
HOW TIOBEPXHOCTH, 00beEMA TIOP M KOHIEHTPALUN
CHJIBHBIX KHUCJIOTHBIX I[EHTPOB I1€0JIUTA, YTO OKA3bI-
BAeT PEIIAIOIee BJAMSHUE HA CEJIEKTHBHOCTD PEAKINN
u ceneKtuBHOCTD N0 n-MIIT .MakcumanbHas ceJiek-
tusHocTb 110 #-UIIT (76.2%) pocruraercs: Ha 1eoJu-
Te, cofepskaiieM 5.0% mac. Mg u 3.0% mac. P.

Kaiouesvle caoea: anKNINpOBaHNE; M30MPOMNa-
HOJI, MaTHUI; napa-ceJeKTUBHOCTD; TOIyoJ; oc-

dop; meoaut ZSM-5.

[ara moctymaenus 04.06.21

In the temperature range 250—350 °C, the effect
of the nature and concentration of modifiers (Mg,
P) on the textural, acidic and catalytic properties
of a zeolite of the ZSM-5 type in the alkylation of
toluene with isopropanol has been studied. On
the unmodified HZSM-5 catalyst, the alkylation
proceeds nonselectively. At 330—350 °C, the
isopropyltoluene (4-IPT) content in the
isopropyltoluene (IPT) mixture is 34.3—44.2 %.
Magnesium oxide plays an important role in the
selectivity of the reaction products and in the
distribution of isopropyltoluene isomers.
Modification of the HZSM-5 zeolite with
magnesium increases the selectivity for IPT to
69.6 wt % and a sharp increase in the content of 4-
IPT in the IPT mixture to 70.8%. Additional
modification of the catalyst with 5% Mg-HZSM-5
with ammonium hydrogen phosphate significantly
affects its catalytic properties. The distribution of
isopropyltoluenes depends on the content and
concentration of phosphorus in the catalyst. An
increase in the phosphorus concentration in the
composition of 5% Mg-HZSM-5 to 3.0 wt.%
Leads to a significant decrease in by-products, an
increase in IPT selectivity (73.5—76.0 %), as well
as an increase in the content of 4-IPT in the
mixture. IPT up to 76.2%. A further increase in
the phosphorus content in 5% Mg-HZSM-5
significantly reduces the selectivity for IPT.69.6—
72.1 decisive influence on the selectivity of the
reaction products and the selectivity for p-
isopropyltoluene (p-IPT). The maximum
selectivity for p-IPT (76.2%) is achieved on a
zeolite containing 5.0 wt% Mg and 3.0 wt% P.

Key words: alkylation; isopropanol;
magnesium, para-selectivity;  phosphorus;
toluene; zeolite ZSM-5.
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Aromatic hydrocarbons are widely used in
the chemical and petrochemical industries 2.
Among aromatic hydrocarbons, p-substituted
alkylaromatic hydrocarbons occupy a special
place 2. The products of toluene alkylation with
isopropanol, m-and p-isopropyltoluenes (IPT)
are technically important products of organic
synthesis 3-3 Currently, there is no perfect
technology for producing isopropyltoluene
(IPT). Complexes based on AlICl;, TiCl,;, BF;
and phosphoric acid have been proposed as
catalysts %3 on kieselguhr, which do not provide
a high yield of IPT, in particular, 4-IPT in
industrial production. Most catalysts give a
mixture of IPT containing all isomers. The nature
of the catalyst significantly affects the isomeric
composition of IPT obtained by alkylation of
toluene #°. H,SO,, BF3H;PO, and H', a mixture
of 2-TPT, 3-IPT and 4-IPT is obtained % °, where
the content of 2-IPT is 13—40 % wt. The use of
phosphoric acid on kieselguhr leads to a mixture
enriched with 2-IPT (59% wt), and the use of
AICl; gives a mixture enriched with 3-IPT (86%
wt) and practically free of 2-IPT. The acid
catalysts used in industry do not allow obtaining
a mixture of IPT enriched with the p-isomer
(4-T1PT). In addition, the alkylation processes
carried out in the presence of Friedel-Crafts
catalysts have significant disadvantages:
environmental pollution with acid waste,
corrosion of equipment, high consumption and
unregenerability of the catalyst . These
disadvantages have led to the need for the search
for heterogeneous catalysts based on zeolites
that exhibit high activity and selectivity in the
reactions of alkylation, isomerization,
disproportionation, oligomerization of various
organic compounds due to their molecular sieve
and acidic properties 7~ on zeolites of the X, Y
type; beta and mordenite with large pores
required for the synthesis of p-IPT 1213,
Recently, high-silica zeolites of the ZSM-5 type,
due to their unique structure with intersecting
straight and sinusoidal channels, size of input
windows ~55nm and adsorption properties, are
everywhere replacing traditional Friedel-Crafts
catalysts 14 15,

One of the possible ways to change the
catalytic and molecular sieve properties of
zeolites in order to obtain para-selective
catalysts for the alkylation of alkylaromatic
hydrocarbons with monoatomic saturated
alcohols C4—C, is their chemical modification
27,1618 The nature of the modifying metal (Mg,
La, Ho, Yb, Pr) significantly affects the para-
selectivity of a catalyst based on a zeolite of the

ZSM-5 type in the alkylation of toluene with
isopropanol.

In this regard, the goal of this work is to
study the effect of the combined modification of
high-silica zeolite ZSM-5 with magnesium and
phosphorus on its physicochemical and catalytic
properties in the alkylation of toluene with
isopropanol.

Experimental part

The initial H-form of zeolite ZSM-5 was
obtained by ion exchange ZSM-5 (ZAO
Nizhnenovgorodskie sorbents, Russia, SiO,/
Al,O3 = 33) with NH,CI: 5.0 g of ZSM-5 was
refluxed twice in 100 ml of 1.0 M NH,CI
solution for 6 hours, washed with distilled
water, dried at 110 °C and calcined at 550 °C,
respectively, for 4 hours % 2°. Modification of
HZSM-5 with magnesium was carried out by
impregnation using an aqueous solution of
magnesium acetate. Samples of Mg-P-HZSM-5
were obtained by impregnation of zeolite, Mg-
HZSM-5 with an aqueous solution of ammonium
hydrogen phosphate. The content of Mg and P in
the catalysts was 5.0% wt and 1.0—3.0 % wt,
respectively.After impregnation, the samples
were dried for 4h at 110 °C and finally pierced
for 4 h in a muffle furnace at 550 °C.

The acidic properties of the modified zeolites
were studied by the method of thermogrammed
thermal desorption of ammonia 9.

The study of the porous structure of the
samples was carried out by the method of low-
temperature adsorption of nitrogen at 77K on an
ASAP-2010 setup from Micromeritics. Before
measurement, samples (about 220 mg) were
degassed at 250 °C and 1 - 103 Pa for 4 hours.
The specific surface area was calculated using
the BET method.

Catalytic experiments were carried out in a
quartz reactor (inner diameter 1.0 cm and length
10 c¢m), which was placed in a microprocessor
controlled furnace. The loading of the catalyst
was 2.0 g. Before the start of the experiments,
the catalysts were activated in an air atmosphere
at 500 °C for an hour. The reaction was carried
out at atmospheric pressure in the presence of
hydrogen in the temperature range 300—400 °C
with a volumetric feed rate of 1 h'! and a molar
ratio of C;Hg: i-CsH;OH : H,=2:1:17.

The reaction products were analyzed on an
Agilent HP gas chromatograph using a capillary
column (100 m x 250 mm) with flame ionization
detector (FID). The analysis was carried out in
the temperature programming mode from 500 to
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250 °C with a heating rate of 10 °C per minute.
The selectivity and product yield were
calculated from the results of gas
chromatographic analysis.

Results and its discussion

Unmodified HZSM-5 zeolite catalyst

Table 1 show the effect of reaction
temperature on catalytic performance of HZSM-
5 catalyst for alkylation of toluene with
isopropanol. From table 1, it can be seen that the
product, contains benzene, isopropyltoluene
(IPT) mixture, propyltoluene mixture, liquid
aliphatic hydrocarbons Cs; and alkylaromatic
hydrocarbons Cg..With an increase in the
reaction temperature to 330 °C, the yield of by-
products increases and the content of 4-IPT in
the mixture of IPT decreases from 44.2% to
34.3%,isomerization of p-cymene to m- and o-
IPT occurs.

Modified HZSM-5 zeolite catalyst

1.Magnesium modified HZSM-5 zeolite
catalyst.

The results of the alkylation of toluene with
isopropanol using 5% Mg- HZSM-5 as a catalyst
are presented in Table 1. As can be seen from

Table 1, the introduction of magnesium with
HZSM-5 impregnation with a solution of
Mg(CH3COO0),, followed by decomposition of
the salt at 550 °C leads to a significant change in
the catalytic properties of the catalysts. The
magnesium oxide has an essential on the
selectivity of the reaction products and the
distribution of isopropyltoluene isomers.

Modification of zeolite HZSM-5 with
magnesium promotes an increase in the
selectivity for IPT to 69.6% wt and a sharp
increase in the content of p-IPT in the IPT
mixture up to 70.8%.

2. Phosphorous and magnesium modified
HZSM-5 zeolite catalyst.

The effect of temperature and phosphorus
concentration on the catalytic properties of the
5% Mg-HZSM-5 catalyst is shown in Table 2.
From the data obtained, it can be seen that the
introduction of 1.0% wt phosphorus into the 5%
Mg-H-ZSM-5 composition significantly affects on
its catalytic properties. For example, at 300 °C,
the conversion of toluene decreases from 17.4 to
15.2 %, and the selectivity to 4-IPT increases
from 70.8 to 72.3 %. An increase in the
phosphorus concentration in the composition of
3% Mg-HZSM-5 to 3.0% wt leads to a signifi-

Table 1

The composition of the alkylation products of toluene with isopropanol
on HZSM-5 and 5% Mg-HZSM-5

— . ,
¢ Conve\:\';atlon, & Selectivity for products in catalyst, % Isc,iir:;':' rl; ICS.TFLZSI_
Catalyst o - I Coal ARCC
toluene 'i‘;ﬁ’]g' benzene| IPT progﬁteo u- Cafgsh';’a_ andhighser 34PT| 4-IPT | 2-IPT
250 | 9.2 95 04 | 524 336 12.2 1.4 506 | 442 | 52
HZSM-5 [ 300 | 19.8 | 994 | 22 | 540 | 30.4 86 438 236 | 401 | 63
330 | 27.7 | 99.9 27 | 558 | 298 6.1 56 573 | 343 | 7.9
5% Mg 300 | 141 | o4 14 | 66.8 | 238 6.4 16 242 | 708 | -
Coons 330 | 218 | 980 17 | 67.9| 215 57 3.2 27| 673 | -
350 | 247 | 98.8 19 | 696 | 19.4 53 3.4 380 | 612 | 0.8
Table 2

The effect of temperature and phosphorous content in the composition of 5% Mg-HZSM-5
on the composition of the products of alkylation of toluene with isopropanol

Content Conversion % Cs. aliph.

atP, |t°C Benzene | IPT | FroPY- 5hydr2 ARHCs | 5 o7 | 4cipT | 2-1PT

% wt toluene | alcohol toluene carbons and higher
300 15.6 91.2 1.2 68.4 22.9 6.1 1.4 277 | 723 -

1.0 330 19.5 95.4 1.5 69.4 20.7 5.5 2.9 309 | 69.1 -
350 | 23.1 97.1 1.7 71.3 18.8 5.1 3.1 36.2 | 63.2 0.6
300 13.8 89.6 0.9 69.9 21.1 5.4 1.1 258 | 74.2 -

2.0 330 17.7 93.7 1.1 71.7 19.9 4.9 24 294 | 70.6 -
350 | 21.6 95.1 1.4 73.1 17.9 4.7 2.8 346 | 654 -
300 12.4 86.8 0.3 73.5 20.6 4.8 0.8 238 | 76.2 -

3.0 330 15.6 90.7 0.5 74.3 19.3 4.4 1.5 279 | 721 -
350 | 20.1 93.6 0.9 76.0 171 4.0 2.0 222 | 67.8 -
300 7.4 79.8 0.2 69.6 214 6.9 0.9 229 | 771 -

4.0 330 10.9 84.6 0.6 71.1 20.7 6.3 1.3 282 | 71.8 -
350 14.8 88.3 0.8 72.1 19.8 5.6 1.7 227 | 673 -
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cant decrease in by-products, an increase in IPT
selectivity (73.5-76.0 %), as well as an increase
in the content of 4-IPT in the mixture IPT up to
76.2%. A further increase in the phosphorus
content in 5% Mg-HZSM-5 significantly
decreases the selectivity for IPT (69.6—72.1 %)
and slightly increases the selectivity for 4-1PT
(77.1%). A sharp decrease in conversion is
observed on a catalyst containing 4% phosphorus
toluene. For example, at 300 °C, the conversion
of toluene is reduced to 7.4%.

The manifestation of the para-selectivity of
catalysts, depending on the nature and
concentration of modifiers in the HZSM-5, can
be due to a change in the acidic and textural
properties of the catalysts as a result of chemical
modification.

As can be seen from Table 3, when
modifying the HZSM-5 zeolite with magnesium
in an amount of 5.0% wt, the concentration of
strong acid sites decreases from 542 umol/g to
227542 umol/g. An increase in the phosphorus
concentration in the 5% Mg-HZSM-5 catalyst
from 1.0% wt to 4.0% wt leads to a gradual
decrease in the concentration of strong acid sites.
The greatest decrease in the concentration of
strong acid sites is achieved on a sample containing
4.0% wt phosphorus and is 89 umol /g. Apparently
for this reason, a decrease in the conversion of
toluene with an increase in the phosphorus
content in the catalysts is associated.

Table 3
Acid characteristics of modified catalysts

Catalyst Tmax, C of the | Concentration
maximum peak of | of acid sites,
ammonia desorption umol/g

I Il C1 C2

HZSM-5 198 418 628 | 542
5% Mg-HZSM-5 188 314 347 | 227
5% Mg-1%P-HZSM-5 187 286 324 192
5% Mg-2%P-HZSM-5 185 274 282 164
5% Mg-3%P-HZSM-5 183 261 228 127

5% Mg-4%P-HZSM-5 180 242 197 89

C; and C, — concentration of acid sites in forms (1)
and (I1).
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Table 4 shows the effect of modification on
the textural characteristics of the catalysts. It is
seen. that with an increase in the concentration
of the modifier, the specific surface area and pore
volume of the zeolite decrease, which indicates a
partial blocking of the zeolite channels by
modifiers, which leads to a change in the size of the
channels and the entrance windows from them.

Thus, the para-selectivity of the modified
catalysts in the alkylation of toluene with
isopropanol with respect to 4-IPT is due to a
significant decrease in the concentration of
strong acid sites and a decrease in the pore
volume of the zeolite as a result of modification.

Table 4
Textural characteristics of modified catalysts

Catalyst SeeT, (M/g) | Vpore, (sm”/g)
HZSM-5 288 0.24
5%Mg- H-ZSM-5 253 0.20
5%Mg-1%P-HZSM-5 238 0.20
5% Mg-2 %P-HZ SM-5 238 0.19
5% Mg-3 %P-HZ SM-5 221 0.17
5% Mg-4 %P-HZ SM-5 194 0.17

Sper — BET specific surface area;
V pore — total pore volume

The introduction of magnesium and
phosphorus by impregnation of HZSM-5 with a
solution of magnesium acetate and ammonium
hydrogen phosphate, followed by calcination at
500 °C leads to a significant change in the acidic,
textural and catalytic properties of the catalysts.
Modification and calcination of the zeolite leads
to the localization of modifiers in micropores and
their distribution on the surface of the zeolite,
where, when interacting with surface OH groups,
they reduce the strength and concentration of
strong acid sites, and also increase the
concentration of acid sites of a more moderate
strength. Catalyst 5% Mg-2% P-HZSM-5
demonstrates high selectivity for 4-1PT (76.2%).
Apparently the modifying effect of Mg and P is
associated with a decrease in the concentration of
strong acid sites and the formation of acid sites of
moderate strength on the catalyst surface and a
change in its porous structure.
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